Author's response to reviews

Title: A culturally adapted lifestyle intervention for a Middle Eastern immigrant population at risk of diabetes: The MEDIM (impact of Migration and Ethnicity on Diabetes In Malmo) intervention study protocol

Authors:

Sanjib Saha (sanjib.saha@med.lu.se)
Matti Leijion (matti.leijion@med.lu.se)
Ulf Gerdtham (ulf.gerdtham@med.lu.se)
Kristina Sundquist (kristina.sundquist@med.lu.se)
Jan Sundquist (jan.sundquist@med.lu.se)
Daniel Arvidsson (daniel.arvidsson@med.lu.se)
Louise Bennet (louise.bennet@med.lu.se)

Version: 2 Date: 7 March 2013

Author's response to reviews: see over